Author's response to reviews

Title: The development of a lay health worker delivered collaborative community based intervention for people with schizophrenia in India

Authors:

    Madhumitha Balaji (madhs@hotmail.com)
    Sudipto Chatterjee (sudipto_dr@yahoo.com.au)
    Mirja Koschorke (mirja.koschorke@lshtm.ac.uk)
    Thara Rangaswamy (thara@scarfindia.org)
    Animish Chavan (animish.chavan7@gmail.com)
    Hamid Dabholkar (hamid.dabholkar@gmail.com)
    Lilly Dakshin (dakshinlilly@gmail.com)
    Pratheesh Kumar (kkpratheesh@gmail.com)
    Sujit John (sujitjohn@scarfindia.org)
    Graham Thornicroft (graham.thornicroft@kcl.ac.uk)
    Vikram Patel (vikram.patel@lshtm.ac.uk)

Version: 2  Date: 7 October 2011

Author's response to reviews: see over
October 6th, 2011

To: Natalie Pafitis  
Executive Editor  
BioMed Central

Dear Natalie

I am re-submitting the paper “The development of a lay health worker delivered collaborative community based intervention for people with schizophrenia in India” (Article no: 6769778485403555) for consideration for publication in BMC Health Services Research. This paper is now being submitted as a correspondence article (previously, this was submitted as “research”; however, please note that as the website does not give “correspondence article” as an option under “article type”, we have chosen the option “research article”). Following your suggestions, we have now removed the detailed results of the formative qualitative phase from the paper and report the results only briefly. The qualitative data will shortly be submitted to the journal separately for publication as an original research article.

This submission contains this covering letter, a title page, the abstract, the text, 3 figures and 1 table. I look forward to the decision of the Journal.

Regards

[Signature]

Professor Vikram Patel  
Sangath Centre, 841/1 Alto Porvorim, Goa 403521, INDIA  
Email: vikram.patel@lshtm.ac.uk  
Tel: +91-9822132038